Author's response to reviews

Title: Aerobic exercise and intraocular pressure in normotensive and glaucoma patients.

Authors:

Konstantinos Natsis (natsis@med.auth.gr)
Irene Asouhidou (iasouhidou@aol.com)
George Nousios (geornous@hotmail.com)
Theodosios Chatzibalis (ctheodos@auth.gr)
Konstantinos Vlasis (kostasvlasis@gmail.com)
Vasilios Karabatakis (karophth@gmail.com)

Version: 5 Date: 27 July 2009

Author's response to reviews: see over
Dear Luciano Quaranta,

I am very sorry that I did not address all the corrections regarding your comments. Thank you very much for your valuable suggestions for improving my manuscript. I have addressed all the issues that have been posed below.

Firstly, the device that was used for assessing BP and HR was the Critikon/Dianamap (Vital Signs Monitor 8100T) and I also mentioned it in the section of Methods.

Secondly, according to our calculation a number of at least 15 patients per group were needed in order to achieve a power of 90% with an a-level of 0.05.

There has been also some corrections in minor spelling errors and re-arrangement of certain paragraphs.

Sincerely
Asouhidou Irene, MD, PhD